Author’s response to reviews

Title: Pragmatic controlled trial to prevent childhood obesity in maternity and child health care clinics: pregnancy and infant weight outcomes (The VACOPP Study)

Authors:

Taina Mustila (taina.mustila@fimnet.fi)
Jani Raitanen (jani.raitanen@uta.fi)
Päivi Keskinen (paivi.keskinen@uta.fi)
Antti Saari (antti.saari@kuh.fi)
Riitta Luoto (riitta.luoto@uta.fi)

Version: 2 Date: 2 February 2013

Author’s response to reviews:

Dear Editor-in-Chief,

You requested the clinical trial number at the end of the abstract. Our trial registration number is now available also at the end of the abstract-text. Please let us now, if there is still something to change.

Sincerely,
Taina Mustila M.D.
Seinäjoki Central Hospital, Hanneksenrinne 7, 60220 Seinäjoki, Finland,
taina.mustila@fimnet.fi
tel. +356503433622